Author's response to reviews

Title: A Cross-sectional Study of the Identification of Prevalent Asthma and Chronic Obstructive Pulmonary Disease among Initiators of Long-Acting beta-Agonists in Health Insurance Claims Data

Authors:

David D Dore (david_dore@brown.edu)
Najat Ziyadeh (najat.ziyadeh@optum.com)
Bin Cai (bin.cai@novartis.com)
Robin Clifford (robin.clifford@optum.com)
Heather Norman (heather.norman@optum.com)
John D Seeger (jseeger@partners.org)

Version: 2 Date: 20 September 2013

Author's response to reviews:

See attached.